**Author's response to reviews**

**Title:** Pediatric reference intervals for thyroid hormone levels from birth to adulthood: a retrospective study

**Authors:**

Klaus Kapelari (klaus.kapelari@i-med.ac.at)  
Christine Kirchlechner (KirchlechnerChristine@gmx.net)  
Wolfgang Hogler (wolfgang.hoegler@i-med.ac.at)  
Katharina Schweitzer (katharina.schweitzer@wienkav.at)  
Irene Virgolini (Irene.virgolini@i-med.ac.at)  
Roy Moncayo (Roy.Moncayo@i-med.ac.at)

**Version:** 5  **Date:** 18 November 2008

**Author's response to reviews:**

The changes given in the e-mail have been done.

1. AUSTRIA > Austria

2. Initials for the e-mails.

3. Manuscript order

4. Shading has been taken out of the tables. Please note that not all tables can be fitted into portrait format. This literature comparison was asked by the reviewers.

5. Table titles down to 1 sentence.

6. Figure 2 has been splitted into 2 and 3, old figure 3 is now 4. The numbers have been changes accordingly in the text.

7. Typographical errors have been looked at.

8. The additional files have been deleted as they were only needed for the reviewers.